Author's response to reviews

Title: The CHOICES/OPCIONES project: Pairing a patient decision aid and patient navigation to improve colorectal cancer screening in vulnerable populations

Authors:

Alison T Brenner (alison.brenner@unc.edu)
Christina M Getrich (chgetrich@salud.unm.edu)
Michael P Pignone (michael_pignone@med.unc.edu)
Robert Rhyne (rrhyne@salud.unm.edu)
Richard M Hoffman (rhoffman@unm.edu)
Andrew McWilliams (andrew.mcwilliams@carolinashealthcare.org)
Brisa Urquieta de Hernandez (brisa.hernandez@carolinashealthcare.edu)
Mark Weaver (mark_weaver@med.unc.edu)
Hazel Tapp (hazel.tapp@carolinashealthcare.org)
Khalil Harbi (harbi@email.unc.edu)
Daniel S Reuland (daniel_reuland@med.unc.edu)

Version: 3
Date: 17 April 2014

Author's response to reviews: see over
April 15, 2014

Dear Editor,

On behalf of my co-authors, I would like to submit our protocol paper entitled “The CHOICES/OPCIONES Study: A randomized controlled trial to compare a decision aid plus patient navigation with usual care on colorectal cancer screening completion in vulnerable populations” for consideration to Implementation Science. It is not being considered for publication anywhere else.

The CHOICES/OPCIONES Study is and American Cancer Society funded randomized controlled trial (American Cancer Society, Grant RSG-13-165-01–CPPB, Reuland PI). The trial will compare an intervention combing patient directed decision aid and patient navigation with usual care for colorectal cancer screening in vulnerable populations, with a secondary focus specifically on Latinos. Clinic sites are in Charlotte, North Carolina and Albuquerque, New Mexico, chosen for their diverse patient populations.

As the corresponding author, I take responsibility for this paper. We hope that you find this protocol paper interesting and will consider publication in Implementation Science.

Sincerely,

Alison Brenner, PhD
Postdoctoral Research Fellow
Cecil Sheps Center for Health Services Research